

Article



Towards Fake News Detection: A Multivocal Literature Review of Credibility Factors in Online News Stories and Analysis Using Analytical Hierarchical Process

Muhammad Faisal Abrar ¹, Muhammad Sohail Khan ², Inayat Khan ³, Mohammed ElAffendi ⁴

- ¹ Department of Computer Software Engineering, University of Engineering and Technology, Peshawar 25120, Pakistan; abrarfaisal49@gmail.com
- ² Department of Computer Software Engineering, University of Engineering and Technology, Mardan 23200, Pakistan; sohail.khan@uetmardan.edu.pk
- ³ Department of Computer Science, University of Engineering and Technology, Mardan 23200, Pakistan; inayatkhan@uetmardan.edu.pk
- ⁴ EIAS Data Science and Blockchain Lab, College of Computer and Information Sciences, Prince Sultan University, Riyadh 11586, Saudi Arabia; affendi@psu.edu.sa
- * Correspondence: ahmad01.shah@ieee.org

Abstract: Information and communication technologies have grown globally in the past two decades, expanding the reach of news networks. However, the credibility of the information is now in question. Credibility refers to a person's belief in the truth of a subject, and online readers consider various factors to determine whether a source is trustworthy. Credibility significantly impacts public behaviour, and less credible news spreads faster due to people's interest in emotions like fear and disgust. This can have negative consequences for individuals and economies. To determine the credibility factors in digital news stories, a Multivocal Literature Review (MLR) was conducted to identify relevant studies in both white and grey literature. A total of 161 primary studies were identified from published (white) literature and 61 were identified from unpublished (grey) literature. As a result, 14 credibility factors were identified, including "number of views", "reporter reputations", "source information", and "impartiality". These factors were then analysed using statistical tests and the Analytic Hierarchy Process (AHP) for decision-making to determine their criticality and importance in different domains.

Keywords: comprehensive synthesis of literature; information trustworthiness; success factors; motivators; digital news stories

1. Introduction

The preservation of information on the World Wide Web (WWW) is known as digital preservation [1]. This is important because the rapid changes in technologies and the short lifespan of digital objects make it challenging to access and preserve information on the WWW for future generations [2,3]. Digital preservation involves various activities, such as preserving the integrity and authenticity of digital objects, ensuring the long-term accessibility of digital information, and managing the risks associated with digital data [4–6]. A critical aspect of digital preservation is the credibility of the information. It is essential to preserve credible information for future generations, as less credible information can have negative consequences, such as disrupting responses to terrorist attacks or natural disasters or affecting stock prices and large-scale investments [8,9]. Several factors can impact the credibility of the information, the accuracy and thoroughness of the information, the clarity of the information, the clarity of the information, and the source's potential biases [10]. It is essential to consider



Citation: Abrar, M.F.; Khan, M.S.; Khan, I.; ElAffendi, M.; Ahmad, S. Towards Fake News Detection: A Multivocal Literature Review of Credibility Factors in Online News Stories and Analysis Using Analytical Hierarchical Process. *Electronics* 2023, *12*, 3280. https:// doi.org/10.3390/electronics12153280

Academic Editor: Ping-Feng Pai

Received: 7 June 2023 Revised: 21 June 2023 Accepted: 7 July 2023 Published: 30 July 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). these factors when evaluating the credibility of the information being preserved [11]. The source of information is a crucial factor to consider when assessing the credibility of digital information. This includes the credibility of the news outlet or publisher and the credibility of the individual or organization providing the information [12]. News outlets and publishers with a history of accuracy and credibility are more likely to present reliable information. It is also essential to consider the credibility of the individual or organization providing the information. The information is more likely to be reliable if the source has a track record of credibility and expertise in the relevant topic. Besides considering the information source, it is also imperative to evaluate the accuracy and thoroughness of the information being preserved [13]. It includes ensuring that the information is supported by evidence and sources and that it is complete and up to date. It is also essential to consider any potential biases in the information, as these can impact its credibility [14]. The clarity of the presentation of the information is another factor to consider when evaluating the credibility of digital information. If the information is clear and understandable, it is more likely to be credible. This includes using appropriate language for the intended audience and ensuring that the information is organized logically and coherently [12,13]. Finally, it is vital to consider the potential biases of the source when evaluating the credibility of digital information. Bias can take many forms, such as political, financial, or personal bias. It is essential to be aware of any potential biases that may impact the credibility of the information being preserved [13,14]. Overall, the credibility of the information being preserved on the WWW is crucial to digital preservation. By considering the source, accuracy and thoroughness, clarity of presentation, and potential biases of the information, it is possible to ensure that only credible information is preserved for future generations [11-14]. Ensuring the reliability and trustworthiness of information accessed by users is crucial in order to avoid negative consequences. This is particularly important for news information, which is often generated from multiple sources in real-time. Thus, it is essential to assess the credibility of news articles based on various criteria, which should be developed systematically. The research aims to identify the factors affecting the credibility of digital news stories, and to address the lack of credibility, it is necessary to distinguish credible information from less credible information through a thorough examination of information diffusion. The study will conduct a multivocal literature review to identify credibility factors for digital news stories. This study is based on the following research questions.

- 1. What factors are identified in the literature (as well as in the grey literature) to be considered for ensuring the digital information credibility of digital news stories?
- 2. How should we rank the identified credibility factors using the Multi-Criteria-Decision-Making (MCDM) algorithm?

This paper is organized into several sections. Section 1 introduces the research topic and objectives. Section 2 reviews existing literature on digital news credibility. Section 3 presents the motivation and novelty. Section 4 explains the methodology, including the research goals, questions, and search strategy. Section 5 presents the results, divided into statistical tests and an analytical hierarchical process. Section 6 discusses implementation challenges and practical insights. Section 7 summarizes the main findings and contributions. Section 8 addresses the limitations and future directions.

2. Literature Review

The increasing digitization of society has made the Internet a central part of many people's lives. In addition to its uses for communication and entertainment, the Internet is an important tool for finding information. The amount of digital information available is continually growing, with more being added daily [15–20]. The Internet has become an essential resource for accessing a vast amount of information on a wide range of topics. It allows users to easily search for and find information on almost any subject they are interested in. This has made it an invaluable resource for students and professionals who need to access reliable and up-to-date information for their work or studies. However, the vast amount of online information can also be overwhelming, and it can be challenging

to determine the credibility and reliability of the sources. It is important for users to be discerning and to carefully evaluate the information they find online to ensure it is accurate and reliable. Overall, the Internet is essential for accessing and sharing information, but it is important to use it responsibly and to critically evaluate the information found online. The individuals involved in education, library services, and app development have a shared interest in disseminating information widely, without distinguishing between credible and less credible sources [21]. While the quality of information is crucial, the focus on promoting credibility is not prioritized as much since it is deemed the responsibility of the end-user [22]. With the prevalence of false information on online platforms, identifying trustworthy sources has become a challenging computational task. Scholars are now shifting their focus towards the nature of information rather than just information itself. This new approach is greatly contributing to research on credible information. Digital information platforms, such as social media, play a significant role in the dissemination of less credible information. These platforms provide information providers with a peer-review-less environment, as highlighted in [23]. However, our focus is on accessing information that plays a minor role in spreading inaccurate information. News, in particular, adds an interesting dimension to this issue. Whether it is political, social, sports, entertainment, or satirical news, it is a major source of interest for people. We suggest that this dimension could be either less credible or more credible. The spreading of news does not see its origin but keeps forwarding it to people, just going through the content at an abstract level. One highlighted reason for spreading news is reading it at a conceptual level. Typically, people do not understand the context and keep forwarding. If we specify the broad terminology of information, it will be news. The term news is considered fascinating in the reader's eye. In order to address the issue of credibility in news stories, it is necessary to differentiate between more and less credible information. To achieve this, it is important to examine the diffusion of information in-depth after distinguishing between credible and less-credible sources. The goal of this study is to identify these factors and their practices. While the increased number of websites has led to a greater volume and variety of available content, the credibility of such content remains a significant question in many situations. The assessment of credibility can be inaccurate if content groups, such as articles, discussion sources, and help, are not properly established [24]. With the rise of Web 2.0 and social media, the internet has become the primary source of information dissemination, with numerous news stories, audio, and videos of varying degrees of credibility. As a result, researchers have shifted their focus to the credibility of content. The terms "credibility" and "trust" are often used interchangeably, with "believability" also being considered an aspect of credibility. People assess credibility by considering multiple dimensions simultaneously, as highlighted by previous research [25–28]. The need for powerful credibility assessment abilities among users is emphasized, as they are the arbiters of accuracy in a domain where they may be informed or uninformed. Assessing information on the web requires awareness of potential bias and vested interests of content writers to determine whether information is appropriate, correct, or plausible. The digital information viewer faces a significant challenge in enhancing the credibility of content, as demonstrated by research from the Stanford University Persuasive Technology (STUP) Lab [9,14,29].

Fog B [29]: Fog B's article delves into the impact of various links and domains on people's perception of credibility and its implications. The research studies discussed in the article focus on the credibility of business and e-commerce websites. The authors in this study emphasize the importance of credibility assessments for web sites, particularly in critical areas such as finance and health. The study aims to identify gaps in consumer education and design guidelines for improving understanding of online credibility. Similarly, when it comes to digital news stories, consumers need to be able to assess the credibility of sources and evaluate the accuracy of the information provided. To accomplish this, they may need to improve their media literacy skills and develop critical thinking strategies. Overall, the study indirectly underscores the importance of credibility assessments not just for websites, but for digital news stories as well.

Maloney, Richard F and Beltramini [23,30,31]: These studies were reported from the literature on the credibility of advertisements in marketing. These research articles analyse the contents and a few attributes of promotion. One of our research's major primary goals is to identify the factors that ensure credibility. In the literature, we found that the perception regarding the advertisement's credibility includes the company's reputation and experience. The credibility of the information is measured via certain factors, i.e., the message contents, source credibility, source bias, and source reputation.

Austin et al. [32]: The study describes a research study that examines the effects of the message type and source reputation on judgments of news believability. The study uses a between-groups 3×3 factorial experiment with a total of 516 participants. The judgments are conceptualized as source credibility and assessments of apparent reality. The study finds that a more innocuous message results in more positive judgments, nor does it interact with the message type. The study concludes that some people base their judgments of news believability more on assessments of the apparent reality of the message content than on the reputation of the media source. Three indices combining measures of source credibility and message apparent reality emerge from a factor analysis, comprising judgments of source truthfulness and message accuracy, source expertise and message representativeness, and source bias and personal perspective.

Wathen et al. [33]: The study discusses how people decide what to believe when seeking information online. It highlights the multidimensionality of the concept of credibility, which is influenced by factors related to source, message, and receiver. However, there is a relative lack of high-quality research evidence, leaving key questions unanswered. To advance our understanding of these issues, research should focus on identifying key markers for credibility of online information, exploring the importance of surface characteristics of web-based media, identifying the best analogy for publication authority/credibility to web sites, and examining the accuracy of the proposed model for judging the credibility of web sites and the information presented on them. From a practical perspective, the text recommends designing an "ideal" web site that emphasizes a good interface, professional image, and source credibility. It also suggests taking advantage of interactive properties of the medium and tailoring content to the beliefs and needs of the audience. The study notes that the interactivity of computer applications is constantly evolving, and the use of human-like virtual agents may change the way users seek and use online information in the future. Credibility issues surrounding medical information have gained significant attention, as nearly 50% of internet users seek health-related information online. Given the importance of such information, many researchers have emphasized the need to assess the credibility of health-related information available on the web. In the case of news, credibility can be assessed by identifying less-credible stories, which can be defined as specific articles, including editorials, news reports, exposes, and other intentionally misleading content.

Rubin et al. [34]: This study discusses the importance of distinguishing between truthful and deceptive news reports, and presents a research project aimed at developing an automated approach for deception detection in news verification. The researchers analysed the rhetorical structures and coherence relation patterns of fabricated and authentic news reports, and used a vector space model to cluster the news by discourse feature similarity. The predictive model achieved 63% accuracy on the test set, which is comparable to human lie detection abilities but not significantly better than chance. The researchers note several confounding issues and methodological limitations that require further research, but suggest that a news verification system could improve credibility assessment of digital news sources and increase new media literacy to prevent the spread of fake news.

Metzger et al. [35]: The article discusses the challenges of locating trustworthy information in digitally networked communication environments and the increasing reliance on information available solely or primarily online. The article focuses on the use of cognitive heuristics in credibility evaluation and presents research findings that illustrate the types of heuristics people use when evaluating the credibility and accuracy of online information. The article concludes with a call for further research to better understand the role and influence of cognitive heuristics in credibility evaluation in computer-mediated communication contexts. In summary, the article highlights the need to better understand how people make judgments about the credibility of online information, given the increasing reliance on digital media for information consumption.

Parth Patwa [36]: The paper introduces a dataset called "Fighting an Infodemic: COVID-19 Fake News Dataset" that addresses the problem of fake news and rumours related to COVID-19 on social media. The dataset consists of 10,700 manually annotated social media posts and articles, both real and fake, on COVID-19. The authors benchmark the dataset using machine learning algorithms and achieve a high performance of 93.46% F1-score with Support Vector Machine (SVM). The paper discusses related work in fake news detection, describes the dataset development process, and highlights the challenges associated with identifying and combating fake news. The dataset statistics reveal differences between real and fake news, such as the length of posts. The paper concludes by emphasizing the importance of tackling fake news during the COVID-19 pandemic and provides the dataset and code for further research.

Bilal Al-Ahmad [37]: The paper addresses the issue of misinformation and fake news related to COVID-19 during the pandemic. The authors propose an evolutionary fake news detection method using four models, aiming to reduce symmetrical features and achieve high accuracy. They apply three wrapper feature selection techniques and evaluate the performance on the Koirala dataset and six derived datasets. The proposed model achieves the best accuracy of 75.43% and outperforms traditional classifiers. The authors suggest applying the methodology to other domains and larger datasets for future work.

M Zivkovic [38]: The paper proposes the use of a modified ant lion optimizer (ALO) to address the issue of false news and disinformation during the COVID-19 epidemic. The ALO algorithm, inspired by the trapping technique of ant lions, is applied for feature selection and dimensionality reduction to enhance classification accuracy. Experimental results demonstrate that the proposed ALO-based technique outperforms other modern classifiers in terms of accuracy, providing an effective approach to combat false news related to COVID-19.

William Scott Paka [39]: The paper introduces the task of COVID-19 fake news detection on Twitter and presents the Cross-SEAN model. They collect a labelled dataset of genuine and fake COVID-19-related tweets, along with unlabelled data. The model incorporates tweet text, features, user information, and external knowledge from credible sources. Cross-SEAN outperforms seven state-of-the-art models and is implemented as a Chrome extension, Chrome-SEAN, which flags fake tweets in real-time. Limitations include potential noise in external knowledge and the need for improved robustness and early detection capabilities.

3. Motivation and Novelty

Ensuring the credibility of information is critical, especially in the digital age, where misinformation and fake news are prevalent. Consumers must be able to distinguish between fact and fiction to make informed decisions and opinions. Credible digital news stories can play a significant role in shaping public opinion, which in turn affects political, social, and economic systems. Thus, it is essential to recognize the role that credibility plays in the consumption of digital news stories and take steps to ensure that the information being consumed is accurate and trustworthy. This can promote informed and critical thinking, which ultimately leads to a more engaged and informed society. The credibility of digital news stories can vary, like any other form of information. Thus, it is crucial to evaluate the reliability of news articles carefully to ensure that the information being consumed is accurate and trustworthy. Several factors can impact the credibility of digital news stories, such as the reputation of the news source, the quality of journalism, and the evidence presented to support the claims made in the article. In addition, the presence of

errors or biased language in the article can also diminish its credibility. To fully address the issue of credibility in digital news stories, it is necessary to adopt a systematic approach to evaluate the reliability of the information. This could involve using a set of established criteria to assess the credibility of news articles or developing new criteria specifically tailored to the digital media landscape. It may also be necessary to consider the specific context in which the news is being consumed, as different audiences may have different standards for what they consider to be credible information. Overall, it is important to recognize the role that credibility plays in the consumption of digital news stories and to take steps to ensure that the information being consumed is accurate and trustworthy. This can help promote informed and critical thinking and lead to a more informed and engaged society.

4. Methodology

Multivocal Literature Reviews (MLRs) are a method used to summarize and evaluate research on a particular topic [40–43]. These reviews differ from traditional systematic literature reviews in that they consider both published research and "grey" literature, such as white papers, videos, and online blogs. MLRs are being increasingly utilized in the fields of computer science and software engineering to identify credibility factors in digital news stories [44–46]. The process of conducting an MLR follows guidelines proposed by Abrar et al. [47] and is depicted in Figure 1. By including both formal and informal sources, MLRs provide a more comprehensive understanding of a topic compared to traditional systematic literature reviews, which only consider formally published research. The process of conducting a Multivocal Literature Review (MLR) is similar to that of a typical systematic mapping (SM) or systematic literature review (SLR), with the primary distinction being the inclusion of grey literature and its consideration. The process begins with the planning and design phase, in which the research goal and questions are defined. This is followed by the conduct phase, which includes several steps such as developing a search strategy and selecting sources, creating a systematic map (classification scheme), and conducting the systematic mapping, synthesis, and review. These steps, including the use of search terms, libraries, and quality assessment, are further described in our protocol paper.

For the quality assessment of grey literature, we used the SADACO approach [47] shown in Figure 2.

We adopted this methodology and found 14 credibility factors from white and grey literature. The aim of this study was to investigate the current state-of-the-art credibility factors, using a MLR approach. This method was chosen because a significant amount of information related to the research topic is available in both the formal and grey literature, including technical reports, blogs, and standards that are not typically published in academic sources. Our MLR process was established by following the guidelines put forth by Abrar et al. [47], illustrated in Figures 1 and 2. Although it closely resembles the conventional SM and SLR processes, it sets itself apart by accommodating and managing grey data. The MLR process begins with the planning and design phase, wherein the research objectives and inquiries are formulated. Afterwards, the MLR is conducted through a series of steps, such as defining the search strategy and selecting the sources, creating a systematic map (classification scheme), performing systematic mapping, synthesis, and review. Each of these steps will be explained in greater detail in the ensuing sections.



Figure 1. Multivocal literature review process.



Figure 2. Grey literature quality assessment process SADACO [47].

4.1. Research Goal and Questions

The primary objectives of this research are

- 1. To find out the credibility factors thorough MLR.
- 2. To rank the identified credibility factors using the Analytical Hierarchical Process.

Our study is based on the following research questions that have motivated the work reported in this document:

RQ 1: What factors are identified in the literature (as well as in the grey literature) to be considered for ensuring the digital information credibility of digital news stories?

RQ 2: How should we rank the identified credibility factors using the Multi-Criteria-Decision-Making (MCDM) algorithm?

4.2. Search Strategy

The initial phase of performing an MLR involves the retrieval and curation of appropriate sources for appraisal. This necessitates devising a search strategy, explicated in this section, as well as specifying the inclusion and exclusion criteria and the benchmarks for assessing quality, described in subsequent sections. The search strategy delineates the process of locating pertinent sources and encompasses various stages. Initially, we deduced the search terms by scrutinizing the research questions' keywords from four angles: population, intervention, relevant outcome, and experimental design, as stated below:

Population: digital news stories.

Intervention: credibility factors, characteristics, practices.

Outcomes of relevance: to determine the factors in ensuring the credibility in digital news stories.

Experimental Design: multivocal literature review.

After confirming the suitability of the search terms we generated, we checked their relevance to the topic by conducting searches on both the Google search engine and academic databases, and also identified synonyms that were related. By applying Boolean operators, we combined these synonyms, and then proceeded to create various search combinations using the keywords. Eventually, we arrived at the final search string that we used in our study as shown in Table 1.

The search string was applied to the following digital libraries:

- IEEE XPLORE;
- SCIENCE DIRECT;
- ACM;
- GOOGLE SCHOLAR (SEARCH ENGINE);
- SPRINGER LINK;

The outcome of the above-mentioned research question is further analysed through some statistical and mathematical tests.

S.NO	Database	Search String
1	GOOGLE SCHOLAR	("Information Credibility" OR "information believability" OR "news credibility") AND ("News Stories" OR "Digital News archives") AND ("credibility factors" OR "credibility indicators" OR "Practices" OR "Solutions")
2	ACM	[[All: "information credibility"] OR [All: "digital information believability"] OR [All: "information believability"] OR [All: "news credibility"] OR [All: "news believability"]] AND [[All: "digital news stories"] OR [All: "news stories"] OR [All: "digital news archives"]] AND [[All: "credibility factors"] OR [All: "credibility indicators"] OR [All: "success factors"] OR [All: "practices"] OR [All: "solutions"]]
3	IEEE XPLORE	("All Metadata":"Information Credibility") OR ("All Metadata":"information believability") OR ("All Metadata" :"news credibility") AND ("All Metadata":"News Stories") OR ("All Metadata":"News archives") AND ("All Metadata" :"credibility factors") OR ("All Metadata": indicators") OR ("All Metadata": "information credibility Practices")
4	SCIENCE DIRECT	("Information Credibility" OR "news credibility" OR "news believability") AND ("News Stories" OR "Digital News archives") AND ("credibility factors" OR "credibility indicators" OR "Practices" OR "Solutions")
5	SPRINGER LINK	("Information Credibility" OR "information believability" OR " news credibility") AND ("credibility factors" OR "credibility indicators" OR "Practices" OR "Solutions")

Table 1. Search strings and databases.

4.3. Statistical Tests

The statistical methodologies employed in this study aimed to analyze the significance and correlations of credibility factors with study strategies. Techniques such as Chi-Square tests and ANOVA were utilized to provide statistical evidence and support the findings concerning the relationship between credibility factors and study strategies. To assess the significance of the identified credibility factors shown in Table 2, a Chi-Square analysis was conducted, considering variables such as Methodologies, Source libraries, and Time. The purpose of the Chi-Square test was to determine if there were any statistically significant variations among the different study strategies. The hypotheses examined aimed to establish whether there existed a notable difference among the study strategies employed for a specific credibility factor. The outcomes of the Chi-Square analysis are displayed in Table 3, providing details such as the Pearson Chi-Square value, degrees of freedom (df), and asymptotic significance. The *p*-value approach was employed to evaluate the significance of the hypotheses. By comparing the *p*-value with the predetermined significance level, decisions regarding the acceptance or rejection of the null hypothesis in favor of the alternative hypothesis were made. If the I-value was found to be lower than the significance level, the null hypothesis was rejected. To analyze the occurrences and factors, an Analysis of Variance (ANOVA) test was conducted on the dataset. This statistical test allowed for the assessment of the significance of the variables by comparing the means and variances between different groups. The appropriate null and alternative hypotheses were defined, and the test statistics for the F-statistic were calculated. The ANOVA analysis results, including the Type III Sum of Squares, degrees of freedom, Mean Square, F-value, and significance level, are presented in Table 4. These findings and analysis are discussed in Section 5.

No	CREDIBILITY FACTORS	FREQUENCY	%AGE
1	Number of Views	180	81
2	Reputation of the content creator	125	56
3	Content creator Followers on social media	188	84
4	Impartiality	111	50
5	Frequent Sharing	205	92
6	Number of Likes	166	75
7	Publisher's Reputations	200	90
8	Source	212	95
9	Relevancy of the contents	163	73
10	Reader's feedback	75	33
11	Background Knowledge	145	65
12	content creator's association	219	99
13	belonging to the news place	199	90
14	Latest updates	105	47

 Table 2. Credibility factors and their frequency distribution.

Table 3. Significant credibility factors based on occurrences.

S.NO	CREDIBILITY FACTORS	%AGE
1.	Number of Views	81
2.	Content creator Followers on social media	84
3.	Frequent Sharing	92
4.	Publisher's Reputations	90
5.	Source	95
6.	Relevancy of the contents	73
7.	Content Creator's Association	99
8.	Belonging to the news place	90

Table 4. Statistical analysis results of credibility factors.

Test	Chi-Square Value	Degrees of Freedom (df)	Significance Level
Pearson Chi-Square	130.414 ^a	90	0.003
Likelihood Ratio	76.843	90	0.837
Linear-by-Linear Association	0.198	1	0.657
N of Valid Cases	2703		

 \overline{a} one cell (0.9%) has an expected count of less than 5. The minimum expected count is 3.84.

4.4. Applying Analytical Hierarchical Process (AHP)

In the second phase of the study, the analytical hierarchy process (AHP) was applied to rank the listed credibility factors and their respective categories. AHP is a multi-criteria decision-making technique developed by Saaty [48] and is known for being a precise and accurate method for ranking and prioritizing items. It has also been used in other research areas to solve complex decision-making problems [49]. The steps involved in implementing AHP are shown in Figure 3. One of the advantages of using AHP in this study is that it is well-suited for analyzing data obtained through the survey process. The steps depicted in Figure 3 are explained in the following sections.



Figure 3. Analytical Hierarchical Process (AHP) stages.

5. Results

During the data extraction phase, we carefully selected a range of literature sources that were relevant to our research goals. These sources included both grey literature, such as reports and surveys, and white literature, such as journal articles and academic papers. We used a pre-defined data extraction form to carefully extract the necessary data from these sources in a standardized and organized manner. The data extraction form was designed to capture a variety of important details about each literature source. This included the date of review, the title of the publication, and the names of the authors. We also recorded the reference for each source and the database from which it was obtained. Additionally, we recorded various credibility factors that have a positive impact on the trustworthiness of the information, such as the methodology used in the study and the publication quality. We also recorded information about the target population and sample population of each study, as well as the type of organization conducting the analysis. This could include media cells, news agencies, or research institutes. We noted the size of the company, as well as the country or location of the analysis. We recorded the year of publication and the type of news covered, such as political, sports, or cultural topics. Finally, we documented whether the news was domestic or foreign in nature and the medium through which it was disseminated, such as television, blogs, or social media. The data were extracted on the pre-defined extraction form from each of the finally selected sources. Our final selection

includes a sample size of 161 papers from grey literature and 61 from white literature. After the data extraction phase, the data synthesis was performed to identify the credibility factors from the extracted data. The data synthesis phase was conducted by the primary reviewer (the primary author) with the help of a secondary reviewer (the co-author). After a thorough review with the Research Evaluation team, we have identified 14 credibility factors from all sources. Table 2 contains list of the credibility factors found from the literature adopting a multivocal literature review.

The debate on some of the credibility factors are discussed below;

Sources : The "source" of a digital news story refers to the origin of the information contained in the story [50]. It can refer to the person or organization that provided the information, as well as the media outlet or platform that published the story. The credibility of the source is a crucial factor in assessing the credibility of a digital news story because it determines the reliability and accuracy of the information presented [51]. If the source is credible, meaning that it has a proven track record of providing accurate and unbiased information, the news story is more likely to be credible. On the other hand, if the source is unreliable or has a history of publishing inaccurate or biased information, the news story may be less credible or even misleading [52]. In the digital age, where anyone can publish news or information online, it is important to be critical of sources and to verify information before accepting it as true. This is especially important in the context of social media, where news stories can spread quickly and without any fact-checking or editorial oversight. By paying attention to the source of a digital news story, readers can better assess its credibility and make informed decisions about whether or not to believe it [50–52].

Number of Views: The number of views that a digital news story receives can impact its perceived credibility in a number of ways [53]. On the one hand, a high number of views can suggest that the story is important or newsworthy, and that many people have found it to be credible and worth sharing with others. This can give the story a certain level of legitimacy and authority, as it has been deemed worthy of attention by a large number of people [54].

Frequent Sharing: Frequent sharing of digital news stories has a positive effect on increasing their reach and visibility, but can also lead to the spread of misinformation and a lack of in-depth understanding [55]. On one hand, the frequent sharing of digital news stories can have a positive effect on their reach and visibility. With the rise of social media and instant messaging apps, news stories can be shared quickly and easily, increasing their potential audience. This can be especially beneficial for stories that may not have received as much attention otherwise. Frequent sharing can also increase the likelihood of a story going viral, resulting in even greater exposure [55–58]. On the other hand, frequent sharing can also have negative consequences. One major issue is the spread of misinformation, as stories are shared without being fact-checked or vetted for accuracy [53]. This can be especially problematic in today's world, where fake news and conspiracy theories can spread quickly and easily online. Additionally, frequent sharing may encourage a "clickbait" mentality, where news outlets prioritize sensational headlines over in-depth reporting and analysis [54,55]. Another issue with frequent sharing is the lack of in-depth understanding that can result. With so many stories being shared on a daily basis, it can be difficult for readers to fully comprehend the context and significance of each individual story. This can lead to a "surface-level" understanding of current events, without fully grasping the nuances or complexities of the issues at hand [54–56].

5.1. Analysis by Applying Statistical Tests

After identifying the credibility factors in digital news stories through MLR, we classified a few factors in Table 3 based on their significance. The criteria for selection of a significant credibility factor as a credibility factor will be considered as a significant credibility factor whose frequency was \geq 70 [16]. The identified considerable credibility factors are "Number of Views", "Content creator Followers on social media", "Frequent Sharing", "Number of Likes", "Publisher's Reputations", "Source", "Relevancy of the

contents", "Content Creator's Association" and "Belonging to the news place." These credibility factors were further evaluated for significance. Hence, we applied Chi-Square statistical analysis based on varinews like methodologies, source libraries, and time. The aim is to recognize whether these credibility factors remain stable/consistent in each methodology, source library, and time, respectively, or vice versa. We used the Chi-Square test to identify statistically significant differences among the various study strategies. The test statistic for the Chi-square is given by the following expression as

$$\chi^2 = \sum_{i=1}^r \sum_{j=1}^c \frac{(o_{ij} - e_{ij})^2}{e_{ij}}$$

In the context of a contingency table, o_{ij} refers to the observed frequency, representing the actual count in the ith row and jth column. Conversely, e_{ij} represents the expected frequency, which is the anticipated count in the *i*th row and *j*th column assuming that the row and column variables are independent (as stated by the null hypothesis). Additionally, 'r' signifies the total number of rows present in the contingency table, while 'c' signifies the total number of columns. To evaluate the significance of the credibility factors, we conducted a Chi-Square statistical analysis based on variables such as methodologies, source libraries, and time. The aim was to determine if these credibility factors remain consistent within each methodology, source library, and time, or vice versa. The Chi-Square test was employed to identify statistically significant differences among the study strategies. Null and alternative hypotheses were examined for the credibility factors as follows:

H0. There is no significant difference among the various study strategies used for a particular credibility factor.

H1. *There is a significant difference among the various study strategies used for a particular motivator.*

A significance level of 0.05 (or 5%) was employed in the analysis, as *p*-values below this threshold are generally regarded as statistically significant. We have used the "SPSS" software for the analysis of these data. The results are presented in Table 4.

Concept of *p***-Value approach:** *p*-value is the minimum significance level value based on which we want to reject Ho. If the *p*-value is greater than the significance level, the null hypothesis cannot be rejected, while the alternative hypothesis is accepted when the *p*-value is less than the significance level. The results of our data are exhibited in Table 4.

The chi-square test was performed on the credibility factors of the variable methodologies, i.e., Systematic Literature Review, Ordinary Literature Review, Case Study, Survey, and Interview. The results shown in Table 4 illustrate that the *p*-value is less than the significance level. Hence, the null hypothesis is rejected, and it concluded that there is a significant association between the factors and the study strategies. Further, we performed the ANOVA test on our data set. The result is illustrated in Table 5. The appropriate null and alternative hypotheses are as follows:

 \mathbf{H}_{0}^{\prime} . The difference between the column mean is zero, i.e., the occurrences are all equal.

 H'_1 . Not all column means are equal.

 $\mathbf{H}_{0}^{\prime\prime}$. The row means are equal, or the factors are equal.

- $\mathbf{H}_{1}^{\prime\prime}$. Not all the factors have equal means.
- $\mathbf{H}_{0}^{\prime\prime\prime}$. There is no interaction between the column and rows.
- $\mathbf{H}_{1}^{\prime\prime\prime}$. The interaction effect is not equal to zero.

The test statistic in this case will be used as: F = Estimated variance from "Between SS"/Estimated variance from "Error SS"

$$F = \frac{{s_1}^2}{{s_3}^2}$$
 and $F = \frac{{s_2}^2}{{s_3}^2}$

where S_1^2 and S_2^2 derived from the "between column means SS," and the "between row means SS". And S_3^2 is derived from errors sum of square.

Source	Hypothesis	Type III Sum of Squares	Degrees of Freedom (df)	Mean Square	F	Sig.
Intercept	Differences in Means	65,234.009	1	65,234.009	294.619	0.000
Error		3321.277	15	221.418 ^a		
Columns	Differences in Means	20,324.179	6	3387.363	124.254	0.000
Error		2453.536	90	27.262 ^b		
Factors	Differences in Means	3321.277	15	221.418	8.122	0.000
Error		2453.536	90	27.262 ^b		
Columns * Factors	Differences in Means	2453.536	90	27.262		
Error		0.000	0	. <i>c</i>		

Table 5. Analysis of Variance (ANOVA) for occurrences and factors.

(Note': ^a MS(Factors), ^b MS (Columns * Factors), ^c MS(Error)).

The *p*-value shown in Table 5 illustrates that the computed values of the F-statistic fall in the critical regions. Hence, we reject all three null hypotheses and accept the three alternative hypotheses. The term "Type III" in the ANOVA table refers to the method used to calculate the sum of squares. It takes into account the presence of other factors in the model when assessing the variation attributed to each source. This allows for a more accurate evaluation of the individual contributions of the factors.

5.2. Analysis by Applying Analytical Hierarchical Process (AHP)

Analytical hierarchical processing (AHP) is a well-known method used for Multi-Criteria Decisions implemented by Satty [48]. The AHP, or Analytic Hierarchy Process, is a method used to help solve complex decision-making problems that have both quantitative and qualitative aspects. It has been widely studied and applied by researchers in a variety of fields [59–62]. Furthermore, Analytical Hierarchical Process is used to prioritize the credibility factors based on their relative importance. The AHP has two phases: the categorization of the factors and applying the AHP. The second phase comprises seven steps that are elaborated on in the subsequent sections.

5.2.1. The Categorization of the Factors

In Phase 1, the credibility factors are mapped to their respective categories through experts. In this study, we have applied the MLR to identify credibility factors. Through our MLR, we identified 14 credibility factors in digital news stories, as shown in Table 2. The credibility factors listed in Table 2 were further classified into three categories: "Content Creator," "Content Background," and "Media Response." Figure 4 shows the mapping of each credibility factor to its respective category according to experts' point of view. The "Content Creator" category is comprised of "Reputation of content creator", "Content creator Followers on social media," "Background Knowledge", "content creator's association"

and "belonging to the news place." The "Content Background" category consists of "Publisher's Reputations", "Source", "Relevancy of the contents," and "Latest updates". The "Media Response" category consists of "Number of views", "Frequent Sharing, "Number of Likes", "Reader's feedback" and "Impartiality". These details are depicted in Figure 4.



Figure 4. The hierarchical structure of credibility.

5.2.2. Decompose the Complex Problem into Its Hierarchical Structure

This step involves identifying goals and categories and ranking the credibility factors. The hierarchical structure of the problem has at least three levels, as illustrated in Figure 5. At the highest level (level 1), the goal of the problem is stated. The factors and subfactors are organized at level 2 and level 3, respectively. We created a hierarchical structure, depicted in Figure 6, based on the categorization shown in Figure 4. This structure maps the credibility factors to their relevant category according to expert opinions. The first level represents the goal of the study, the second level represents the categories of credibility factors, and the third level represents the credibility factors (subfactors).

IN DIGITAL **NEWS STORIES**

CONTENT CREATOR

- Reputation of content creator
- Content creator Followers on social media
- Background knowledge
- Content creator's association
- Belonging to the news place

CONTENT BACKGROUND

- Publisher's Reputations
- Relevancy of the contents

- Impartiality



Figure 5. The hierarchical structure of the decision problem.



Figure 6. Categorization and mapping of credibility factors.

5.2.3. Construct a Pair-Wise Matrix of Sub-Factors to Find the Priority Weight/Vector of Credibility Factors

To apply the analytical hierarchy process (AHP) to prioritize the credibility factors and their categories, a pair-wise comparison survey was conducted. The survey was completed by 43 participants, which may raise concerns about the representatives of the sample and

the potential impact on the study's findings and conclusions. However, AHP is a subjective method that is able to handle small samples of data [6], and it has been used in other studies with relatively small sample sizes. For example, studies by Shameem et al. [49] and Cheng and Li [49] used small sample sizes of only five and nine participants, respectively, to evaluate perceptions and experiences or prioritize factors. Wong and Li [63] also used a sample of nine experts to survey intelligent building systems using the AHP method. Based on these examples, the sample size of 43 responses used in this study appears sufficient for evaluating the data collected through the AHP process. The pair-wise comparison matrix can be explained by comparing two credibility factors, i.e., CF-1 and CF-2, regarding their importance in ensuring credibility in digital news stories. CF-1 is 7 concerning CF-2 because CF-1 has a seven-degree more excellent value than CF-2, as shown in Table 6.

Table 6. Example of pair-wise comparison matrix.

	CF-1	CF-2
CF-1	1	7
CF-2	1/7	1

To determine the relative importance of the credibility factors and their categories, we used the same method to create pair-wise comparison matrices for each. We employed a nine-point standardized comparison scale, as shown in Table 7 and described in Table 8, to assess the significance of each factor and category.

Table 7. Details of the intensity scale.

Description	Significance Intensity
Equally important	1
Moderately important	3
Strongly more important	5
Very strongly more important	7
Extremely more important	9
Intermediate values	2,4,6,8

Table 8. Description of the 9-point scale for the intensity of importance.

Size of Matrix	1	2	3	4	5	6	7	8	9	10
RI	0	0	0.58	0.9	1.12	1.32	1.41	1.45	1.45	1.49

As seen in Table 8, RI represents a Random Index, and its value changes depending on the matrix size. After collecting the survey responses, we developed the pair-wise matrix. Tables 9–12 represent the pair-wise matrix for each category. We used CF in each table, which stands for "Credibility Factor."

	CF2	CF3	CF11	CF12	CF13
CF2	1	3	1/3	1/3	1/5
CF3	1/3	1	1/5	1/5	1/7
CF11	3	5	1	3	1/3
CF12	3	5	1/3	1	1/3
CF13	5	7	3	3	1

Table 9. Pair -wise matrix for the category of "Content Creator".

Table 10. Pair-wise matrix for category of "Content Background".

	CF7	CF8	CF9	CF14
CF7	1	1/7	1/3	1/7
CF8	7	1	7	3
CF9	3	1/7	1	1/3
CF14	7	1/3	3	1

Table 11. Pair-wise matrix for category of "Media Response".

	CF1	CF4	CF5	CF6	CF10
CF1	1	1/3	1/2	1/5	1/7
CF4	3	1	1/3	1/7	1/5
CF5	2	3	1	1/2	1/3
CF6	5	7	2	1	1/2
CF10	7	5	3	2	1

Table 12. Pair -wise matrix for all categories.

	CONTENT CREATOR	CONTENT BACKGROUND	MEDIA RESPONSE
CONTENT CREATOR	1	3	5
CONTENT BACKGROUND	1/3	1	3
MEDIA RESPONSE	1/5	1/3	1

5.2.4. Calculate the Priority Weight of the Credibility Factor

The pair-wise comparison is performed to calculate the factors' priority weights [64]. Based on their relative significance and the criteria specified at the higher level, credibility factors are compared at every level. The pair-wise comparison matrices are used to calculate the priority weight as follows:

- 1. Matrix: pair-wise comparison matrix of the credibility factors. These pair-wise matrices are discussed in the section "construction of pair-wise matrix".
- 2. Normalizing the matrix: divide each value in each column by the sum of that column. These pair-wise matrices given in section "construction of pair-wise matrix" as Tables 9–12 are then passed to another phase where we divide each value by the sum of that column. This process produces the normalized matrices.
- 3. Priority weight: calculate an average of each row of a matrix for normalization.

 λ Max calculation for the Category "**Content Creator**" is given in Table 13:

$$\lambda Max = (12.3333333 * 0.087241046) + (21 * 0.042676137) + (4.866666667 * 0.25018496) + (7.53333333 * 0.169690354) + (2.009524 * 0.450207503) = 5.37277528$$
(1)

$$CI = (Equation (1) - 5)$$

/(5-1) = (5.37277528 - 5)/(5 - 1) (2)
= 0.09319382

$$CR = Equation (2)/1.12 = 0.09319382/1.12$$

= 0.08320877 \le 0.1(Consistency Okay). (3)

 Table 13. Normalized matrix for category of "Content Creator".

	CF2	CF3	CF11	CF12	CF13	Priority Weight
CF2	0.081	0.143	0.068	0.044	0.1	0.087241046
CF3	0.027	0.048	0.041	0.027	0.071	0.042676137
CF11	0.243	0.238	0.205	0.398	0.166	0.25018496
CF12	0.243	0.238	0.068	0.133	0.166	0.169690354
CF13	0.405	0.333	0.616	0.398	0.498	0.450207503

The λ Max calculation for the category "**Content Background**" is given in Table 14:

 $\lambda Max = (18 * 0.051279377) + (1.619047619 * 0.573598943)$ + (11.33333333 * 0.104401335) + (4.476190476 * 0.270720345)= 4.246723749(4)

$$CI = (Equation (4) - 4)/(4 - 1)$$

= (4.246723749 - 4)/(4 - 1) = 0.08224125 (5)

$$CR = Equation (5)/0.9 = (0.08224125)/0.9 = 0.091379166 \le 0.1$$

(Consistency Okay). (6)

Table 14. Normalized matrix for the category of "Content Background".

	CF7	CF8	CF9	CF14	Priority Weight
CF7	0.056	0.088	0.029	0.032	0.051279377
CF8	0.389	0.618	0.618	0.67	0.573598943
CF9	0.167	0.088	0.088	0.074	0.104401335
CF14	0.389	0.206	0.265	0.223	0.270720345

The λ Max calculation for the category **"Media Response"** is given in Table 15:

$$\lambda Max = (18 * 0.053364915) + (16.33333 * 0.081) + (6.83333333 * 0.144882087) + (3.842857143 * 0.297802896) + (2.176190476 * 0.422800085) = 5.340553846$$
(7)

$$CI = (Equation (7) - 5)/(5 - 1) = (5.340553846 - 5)/(5 - 1) = 0.085138461$$
(8)

$$CR = Equation (8)/1.12 = (0.085138461)/1.12$$

= 0.076016483 < 0.1(Consistency Okay). (9)

Table 15. Normalized matrix for the category of "Media Response".

	CF1	CF4	CF5	CF6	CF10	Priority Weight
CF1	0.056	0.02	0.073	0.052	0.066	0.053364915
CF4	0.167	0.061	0.049	0.037	0.092	0.081
CF5	0.111	0.184	0.146	0.13	0.153	0.144882087
CF6	0.278	0.429	0.293	0.26	0.23	0.297802896
CF10	0.389	0.306	0.439	0.52	0.46	0.422800085

A λ Max calculation for **all categories** is given in Table 16:

$$\lambda Max = (1.533 * 0.6333) + (4.333 * 0.2605) + (9.000 * 0.1062) = 3.055361493$$
(10)

$$CI = (Equation (10) - 3)/(3 - 1) = (3.055361493 - 3)/(3 - 1) = 0.027680747$$
(11)

$$CR = Equation (11)/0.58 = 0.027680747/0.58$$

= 0.047725425 < 0.1(Consistency Okay). (12)

Table 16. Normalized matrix for overall categories.

	CONTENT CREATOR	CONTENT BACK- GROUND	MEDIA RESPONSE	PRIORITY
CONTENT CREATOR	0.6522	0.6923	0.5556	0.6333
CONTENT BACKGROUND	0.2174	0.2308	0.3333	0.2605
MEDIA RESPONSE	0.1304	0.0769	0.1111	0.1062

5.2.5. Perform Consistency Check

Priority factors are only acceptable when the Consistency Ratio value is less than 0.1, and Consistency Ratio (CR) values up to 0.1 are acceptable. To enhance the consistency of the pair-wise table, the procedure should be repeated if the CR values are not within the recommended range. To calculate the pair-wise matrix consistency, a consistency ratio

(CR) and consistency index (CI) are used in AHP [49,63]. Using the following equation, we can check the consistency of the pair-wise matrix : $CI = \frac{\lambda \max - n}{n-1}$ Where CI represents the consistency index, λ max is the eigenvalue of the matrix, and n represents the size of the matrix or the number of credibility factors in the matrix. We will find the consistency ratio by the below equation after finding the CI: CR = CI/RI Where CR represents the consistency ratio, CI is used for the consistency index, and RI is the random consistency index illustrated in Table 7, which has constant values. Each credibility factor's weighted value (W) is calculated by averaging the normalized values of the corresponding row shown in Tables 13–16. Hence, the λ Max calculation for each category is given above with the normalized table.

5.2.6. Calculate the Credibility Factor's Local Weight (LW) and Global Weight (GW)

The local weight of a credibility factor is the priority weight assigned to each credibility factor inside its respective category. As a result, all the credibility factors' priority weights relative to their categories are calculated and listed in this stage. The value of the local weight inside each category multiplied by the value of the local weight of the corresponding category produces the value of the global weight of each Credibility Factor. We have calculated both LW and GW, which are displayed in Table 17.

Categories	Categories Weight	Factors	Local Weights	Local Ranking	Global Weights	Final Priority
	0.6333	CF2	0.0872	4	0.0553	6
		CF3	0.0427	5	0.027	10
CONTENT CREATOR		CF11	0.2502	2	0.1585	2
		CF12	0.1697	3	0.1075	4
		CF13	0.4502	1	0.2851	1
	0.2605	CF7	0.0513	4	0.0134	12
		CF8	0.5736	1	0.1494	3
CONTENT BACKGROUND		CF9	0.1044	3	0.0272	9
		CF14	0.2707	2	0.0705	5
		CF1	0.0534	5	0.0057	14
		CF4	0.0812	4	0.0086	13
MEDIA RESPONSE	0.1062	CF5	0.1449	3	0.0154	11
	-	CF6	0.2978	2	0.0316	8
		CF10	0.4228	1	0.0449	7

Table 17. Summary of local and global weights of credibility factors and their rankings.

5.2.7. Identify and Create the Overall Priority Ranking

The final list of credibility factors in digital news stories, based on each credibility factor's global weight, is created in this step. Credibility factors are considered highly ranked if they have a more excellent global weight value across all categories.

The study identified 14 different factors and ranked them based on their global weight, with higher global weight values indicating greater importance. The most important factor was "CF13: Belonging to the news place," while the least important factor was "CF1: Number of views." The findings are summarized in Table 18. These results suggest that the "number of views" a digital news story receives is a lesser importance of its credibility, while the news "Belonging To the News Place" suggested a high credibility. It is important to note that the specific factors and their rankings may vary depending on the specific context and research method used in the study.

S.NO	Credibility Factors	Priority
CF13	Belonging To the News Place	1
CF11	Background Knowledge	2
CF8	Source	3
CF12	Content Creator's Association	4
CF14	Latest Updates	5
CF2	Reputation of Content Creator	6
CF10	Reader's Feedback	7
CF6	Number of Likes	8
CF9	Relevancy of The Contents	9
CF3	Content Creator Followers on social media	10
CF5	Frequent Sharing	11
CF7	Publisher's Reputations	12
CF4	Impartiality	13
CF1	Number of views	14

Table 18. Prioritization of credibility factors.

6. Implementation Challenges

Undertaking a multivocal literature review presents several challenges. Firstly, the identification and selection of pertinent sources from diverse disciplines and perspectives demand careful consideration and a comprehensive understanding of the research topic. Moreover, the diversity of sources, including variations in language, culture, and epistemology, complicates the comparison and synthesis of different perspectives and voices. Integrating these divergent viewpoints, which may entail distinct assumptions, values, and conceptual frameworks, poses an additional challenge. Ensuring the quality and credibility of sources used in the review is also crucial, but can be demanding due to varying levels of reliability and trustworthiness. Finally, analysing and synthesising information obtained from diverse sources necessitates a deep understanding of the research topic and the ability to identify key themes and concepts across multiple perspectives. Implementing the Analytic Hierarchy Process (AHP) brings forth its own array of challenges. Primarily, the subjective nature of pairwise comparisons in AHP introduces the potential for biases and inconsistencies in the decision-making process. Handling the complexity of breaking down intricate decisions into smaller components and assessing their relative importance can be time-consuming and arduous, particularly when dealing with a multitude of criteria and alternatives. Ensuring the quality of data used in AHP analysis is paramount, as flawed or incomplete data can compromise the accuracy and reliability of results. Effective communication among multiple stakeholders is essential to navigate challenges related to collaboration and mitigate potential misunderstandings. Lastly, possessing the requisite technical skills and expertise is critical for the successful implementation of the AHP method, and decision-makers may require adequate training and support to fully comprehend and apply the methodology.

7. Conclusions

The study aimed to identify the factors that contribute to the credibility of digital news stories by analysing both white and grey literature. This research is crucial in addressing the pressing issue of fake news and misinformation, as it provides valuable insights for developing strategies to evaluate and verify the accuracy and reliability of digital news information. We adopted an MLR and identified 14 credibility factors from this literature. They then analysed the results of their research using statistical and mathematical tests.

In order to achieve the goals of this study, we have adopted the multivocal literature review. Systematic mapping (SM) and systematic literature reviews (SLRs) are research methods that are commonly used in the fields of computer science and engineering to summarize the existing knowledge and evidence on a particular topic. These methods involve systematically searching for and reviewing relevant research studies and organizing the results in a structured and systematic way. This study identified and ranked various factors that contribute to the credibility of digital news stories. The study identified 14 different factors and ranked them based on their global weight, with higher global weight values indicating greater importance. The most important factor was "CF13: Belonging to the news place," while the least important factor was "CF1: Number of views." These results suggest that the number of views a digital news story receives is a strong indicator of its credibility, while the news source's reputation is of lesser importance. It is important to note that the specific factors and their rankings may vary depending on the context and research method used. Different studies may identify different factors as being most important, and the relative importance of each factor may vary depending on the specific goals or focus of the study. Additionally, the specific ranking of the factors may be influenced by the method used to calculate the global weight values. Despite these limitations, the study provides valuable insights into the factors that contribute to the credibility of digital news stories and can inform the development of strategies for evaluating and verifying the accuracy and reliability of this information. The multivocal literature review (MLR) allowed us to gather enough evidence to support our conclusions about credibility factors in digital information. The MLR was conducted in a systematic manner and the sources were carefully evaluated using predetermined quality criteria. This thorough approach increases our confidence in the credibility of certain news sources and reduces the risk of invalid conclusions.

8. Limitations and Future Directions

This para focuses on the challenges to validity and how they are resolved to build trust in the study's findings. Missing sources that were published in other databases or after conducting the research represent one of the potential threats. To obtain as much information as possible, we used an MLR to compile all relevant articles that had been published in both formal and informal literature. We searched a total of five electronic databases. This guaranteed thorough coverage of all sources. We, therefore, have enough evidence to conclude that these sources address most information credibility elements. Another drawback is that the credibility factors might occasionally be subjective. Based on our own experiences and the knowledge we obtained from the MLR, we carried out the assignment. The factors were iteratively validated by a team of five researchers from various universities to reduce this threat. The credibility factors that we obtained because of our MLR are a good representation of the actual factors in information sciences. However, because some sources do not provide adequate or clear information relating to our research topics, source selection and data extraction in the MLR are subjective. To address this potential threat, careful screening and selection of sources was conducted based on a set of quality criteria. The quality criteria were designed for both white and grey literature in our previous study.

One potential direction for future research on the topic of digital news credibility could be to expand upon the factors identified in the study and to examine the relationships between these factors and the credibility of digital news stories. This could involve conducting additional research to gather more data on the factors and their impact on credibility and using these data to develop more nuanced and refined models for assessing the credibility of digital news stories. Another potential direction for future research could be to explore the specific strategies and methods that can be used to evaluate and verify the accuracy and reliability of digital news stories. This could involve examining the effectiveness of different approaches to fact-checking or developing new tools and technologies to help people assess the credibility of news articles. Overall, the future direction of research on the topic of digital news credibility will likely be influenced by the continued evolution of the media landscape and the emergence of new challenges and opportunities in the field. By staying abreast of these developments and continuing to explore the factors that contribute to the credibility of digital news stories, it will be possible to better understand and address the challenges of ensuring the accuracy and reliability of information in the digital age.

Author Contributions: Conceptualization, M.F.A. and M.S.K.; methodology, M.F.A. and I.K.; software, S.A.; validation, M.F.A., M.S.K. and S.A.; formal analysis, M.F.A.; investigation, M.S.K; resources, S.A.; data curation, M.S.K.; writing—original draft preparation, M.F.A.; writing—review and editing, M.F.A., M.E. and M.S.K.; supervision, M.S.K.; project administration, M.S.K.; funding acquisition, S.A. and M.E. All authors have read and agreed to the published version of the manuscript.

Funding: This research was supported by the EIAS Data Science and Blockchain Lab, College of Computer and Information Sciences, Prince Sultan University, Riyadh 11586, Saudi Arabia.

Data Availability Statement: The data presented in this study are available on request from the authors.

Acknowledgments: The authors would like to acknowledge Prince Sultan University and EIAS: Data Science and Blockchain Laboratory for their valuable support.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. de Kunder, M. Worldwidewebsize. Available online: http://www.worldwidewebsize.com/ (accessed on 1 January 2020).
- Lavoie, B.F. The open R2archival information system reference model: Introductory guide. *Microform Imaging Rev.* 2004, 33, 68–81. [CrossRef]
- 3. Khan, M.; Rahman, A.U. A systematic approach towards web preservation. Inf. Technol. Libr. 2019, 38, 71–90. [CrossRef]
- 4. Khan, M. Using tExt Processing Techniques for Linking News Stories for Digital Preservation. PhD Thesis, Faculty of Computer Science, Preston University, Kohat, Pakistan, 2018.
- Khan, M.; Ur Rahman, A.; Daud Awan, M. Term-based approach for linking digital news stories. In Proceedings of the Italian Research Conference on Digital Libraries, Udine, Italy, 25–26 January 2018; Springer: Berlin/Heidelberg, Germany, 2018; pp. 127–138.
- Khan, M.; Rahman, A.U.; Awan, M.D.; Alam, S.M. Normalizing digital news-stories for preservation. In Proceedings of the 2016 Eleventh International Conference on Digital Information Management (ICDIM), Porto, Portugal, 19–21 September 2016; IEEE: Piscataway, NJ, USA, 2016; pp. 85–90.
- Khan, M.; Rahman, A.U. Digital news story preservation framework. In *Digital Libraries: Providing Quality Information, Proceedings* of the 17th International Conference on Asia-Pacific Digital Libraries, ICADL, Seoul, Republic of Korea, 9–12 December 2015; Springer: Berlin/Heidelberg, Germany, 2015; p. 350.
- 8. Hawkins, D.T. What is credible information? Online 1999, 23, 86–89.
- Fogg, B.J.; Marshall, J.; Laraki, O.; Osipovich, A.; Varma, C.; Fang, N.; Paul, J.; Rangnekar, A.; Shon, J.; Swani, P.; et al. What makes web sites credible? A report on a large quantitative study. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, Seattle WA, USA, 31 March–5 April 2001; pp. 61–68.
- Gupta, A.; Lamba, H.; Kumaraguru, P.; Joshi, A. Faking sandy: Characterizing and identifying fake images on twitter during hurricane sandy. In Proceedings of the 22nd International Conference on World Wide Web, Rio de Janeiro, Brazil, 13–17 May 2013; pp. 729–736.
- Viviani, M.; Pasi, G. Credibility in social media: Opinions, news, and health information—A survey. Wiley Interdiscip. Rev. Data Min. Knowl. Discov. 2017, 7, e1209. [CrossRef]
- 12. Rapoza, K. Can 'fake news' impact the stock market? Forbes News, 26 February 2017.
- Mendoza, M.; Poblete, B.; Castillo, C. Twitter under crisis: Can we trust what we RT? In Proceedings of the First Workshop on Social Media Analytics, Washington, DC, USA, 25 July 2010; pp. 71–79.
- Starbird, K.; Maddock, J.; Orand, M.; Achterman, P.; Mason, R.M. Rumors, false flags, and digital vigilantes: Misinformation on twitter after the 2013 boston marathon bombing. In Proceedings of the IConference, Berlin, Germany, 1 March 2014.
- 15. Fischer, F.; Dockweiler, C. Qualität von onlinebasierter Gesundheitskommunikation. In *eHealth in Deutschland*; Springer: Berlin, Germany, 2016; pp. 407–419.
- Abrar, M.F.; Khan, M.S.; Ali, S.; Ali, U.; Majeed, M.F.; Ali, A.; Amin, B.; Rasheed, N. Motivators for large-scale agile adoption from management perspective: A systematic literature review. *IEEE Access* 2019, 7, 22660–22674. [CrossRef]
- Faisal Abrar, M.; Sohail, M.; Ali, S.; Faran Majeed, M.; Ali Shah, I.; Rashid, N.; Ullah, N. De-motivators for the adoption of agile methodologies for large-scale software development teams: An SLR from management perspective. *J. Softw. Evol. Process* 2020, 32, e2268. [CrossRef]

- Daud, S.; Ullah, M.; Rehman, A.; Saba, T.; Damaševičius, R.; Sattar, A. Topic Classification of Online News Articles Using Optimized Machine Learning Models. *Computers* 2023, 12, 16. [CrossRef]
- Khan, S.A.; Shahzad, K.; Shabbir, O.; Iqbal, A. Developing a Framework for Fake News Diffusion Control (FNDC) on Digital Media (DM): A Systematic Review 2010–2022. *Sustainability* 2022, 14, 15287. [CrossRef]
- 20. Shahzad, K.; Khan, S.A.; Ahmad, S.; Iqbal, A. A Scoping Review of the Relationship of Big Data Analytics with Context-Based Fake News Detection on Digital Media in Data Age. *Sustainability* **2022**, *14*, 14365. [CrossRef]
- 21. Kapoun, J. Teaching undergrads WEB evaluation. Coll. Res. Libr. News 1998, 59, 522–523. [CrossRef]
- 22. Jadad, A.R.; Gagliardi, A. Rating health information on the Internet: Navigating to knowledge or to Babel? *Jama* **1998**, 279, 611–614. [CrossRef] [PubMed]
- 23. Maloney, J.C. Is advertising believability really important? Mark. Manag. 1994, 3, 47. [CrossRef]
- 24. Warwick, C.; Terras, M.; Galina, I.; Huntington, P.; Pappa, N. Library and information resources and users of digital resources in the humanities. *Program* 2008, 42, 5–27. [CrossRef]
- Fogg, B.J.; Tseng, H. The elements of computer credibility. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, Pittsburgh, PA, USA, 15–20 May 1999; pp. 80–87.
- Fogg, B.J.; Soohoo, C.; Danielson, D.R.; Marable, L.; Stanford, J.; Tauber, E.R. How do users evaluate the credibility of Web sites? A study with over 2,500 participants. In Proceedings of the 2003 Conference on Designing for User Experiences, San Francisco, CA, USA, 6–7 June 2003; pp. 1–15.
- 27. Hovland, C.I.; Weiss, W. The influence of source credibility on communication effectiveness. *Public Opin. Q.* **1951**, *15*, 635–650. [CrossRef]
- Iding, M.K.; Crosby, M.E.; Auernheimer, B.; Barbara Klemm, E. Web site credibility: Why do people believe what they believe? Instr. Sci. 2009, 37, 43–63. [CrossRef]
- Stanford, J.; Tauber, E.R.; Fogg, B.; Marable, L. Experts vs. Online Consumers: A Comparative Credibility Study of Health and Finance Web Sites. 2002. Available online: https://www.semanticscholar.org/paper/Experts-vs.-Online-Consumers%3A-A-Comparative-Study-Stanford-Tauber/32aed7114bd3c2fc9d3e9a6b623033f9e7315f72 (accessed on 1 June 2021).
- 30. Cheskin, R. *Trust in the Wired Americas;* Cheskin Research Report; California Cheskin Research: Redwood Shores , CA, USA, 2000 ; pp. 1–32.
- 31. Beltramini, R.F.; Stafford, E.R. Comprehension and perceived believability of seals of approval information in advertising. *J. Advert.* **1993**, *22*, 3–13. [CrossRef]
- 32. Austin, E.W.; Dong, Q. Source v. content effects on judgments of news believability. Journal. Q. 1994, 71, 973–983. [CrossRef]
- 33. Wathen, C.N.; Burkell, J. Believe it or not: Factors influencing credibility on the Web. J. Am. Soc. Inf. Sci. Technol. 2002, 53, 134–144. [CrossRef]
- Rubin, V.L.; Conroy, N.J.; Chen, Y. Towards news verification: Deception detection methods for news discourse. In Proceedings
 of the Hawaii International Conference on System Sciences, Kauai, HI, USA, 5–8 January 2015; pp. 5–8.
- 35. Metzger, M.J.; Flanagin, A.J. Credibility and trust of information in online environments: The use of cognitive heuristics. *J. Pragmat.* **2013**, *59*, 210–220. [CrossRef]
- 36. Patwa, P.; Sharma, S.; Pykl, S.; Guptha, V.; Kumari, G.; Akhtar, M.S.; Ekbal, A.; Das, A.; Chakraborty, T. Fighting an infodemic: Covid-19 fake news dataset. In Proceedings of the Combating Online Hostile Posts in Regional Languages during Emergency Situation: First International Workshop, CONSTRAINT 2021, Collocated with AAAI 2021, Virtual Event, 8 February 2021; Springer: Berlin, Germany, 2021; pp. 21–29.
- Al-Ahmad, B.; Al-Zoubi, A.; Abu Khurma, R.; Aljarah, I. An evolutionary fake news detection method for COVID-19 pandemic information. *Symmetry* 2021, 13, 1091. [CrossRef]
- Zivkovic, M.; Bacanin, N.; Arandjelovic, J.; Stanojlovic, S.; Rakic, A.; Venkatachalam, K. COVID-19 Fake News Detection by Improved Ant Lion Optimizer Metaheuristics. In *Artificial Intelligence and Sustainable Computing: Proceedings of ICSISCET, Gwalior, India*, 13–14 November 2021; Springer: Berlin, Germany, 2022; pp. 469–484.
- Paka, W.S.; Bansal, R.; Kaushik, A.; Sengupta, S.; Chakraborty, T. Cross-SEAN: A cross-stitch semi-supervised neural attention model for COVID-19 fake news detection. *Appl. Soft Comput.* 2021, 107, 107393. [CrossRef] [PubMed]
- 40. Garousi, V.; Felderer, M.; Mäntylä, M.V. Guidelines for including grey literature and conducting multivocal literature reviews in software engineering. *Inf. Softw. Technol.* **2019**, *106*, 101–121. [CrossRef]
- Fernández, A.; Gómez, B.; Binjaku, K.; Meçe, E.K. Digital transformation initiatives in higher education institutions: A multivocal literature review. In *Education and Information Technologies*; Springer: Berlin, Germany, 2023; pp. 1–32.
- Baysal, M.V.; Özcan-Top, Ö.; Betin-Can, A. Blockchain technology applications in the health domain: A multivocal literature review. J. Supercomput. 2023, 79, 3112–3156. [CrossRef] [PubMed]
- 43. Trendowicz, A.; Groen, E.C.; Henningsen, J.; Siebert, J.; Bartels, N.; Storck, S.; Kuhn, T. User experience key performance indicators for industrial IoT systems: A multivocal literature review. *Digit. Bus.* **2023**, *3*, 100057. [CrossRef]
- Leite, L.; Meirelles, P.R.M.; Kon, F.; Rocha, C. Practices for Managing Machine Learning Products: A Multivocal Literature Review. 2023. Available online: https://www.techrxiv.org/articles/preprint/Practices_for_Managing_Machine_Learning_Products_a_ Multivocal_Literature_Review/21960170 (accessed on 4 June 2023).

- 45. Gramlich, V.; Guggenberger, T.; Principato, M.; Schellinger, B.; Urbach, N. A Multivocal Literature Review of Decentralized Finance. Available online: https://www.researchgate.net/profile/Marc_Principato/publication/368081185_A_Multivocal_ Literature_Review_of_Decentralized_Finance_Current_Knowledge_and_Future_Research_Avenues/links/63dbfa2ec465a873 a27cd992/A-Multivocal-Literature-Review-of-Decentralized-Finance-Current-Knowledge-and-Future-Research-Avenues.pdf (accessed on 1 June 2023).
- Hernández, R.; Moros, B.; Nicolás, J. Requirements management in DevOps environments: A multivocal mapping study. *Requir. Eng.* 2023, 1–30. [CrossRef]
- 47. Abrar, M.F.; Khan, M.S.; Khan, I.; Ali, G.; Shah, S. Digital Information Credibility: Towards a Set of Guidelines for Quality Assessment of Grey Literature in Multivocal Literature Review. *Appl. Sci.* **2023**, *13*, 4483. [CrossRef]
- Abdolvand, N.; Albadvi, A.; Aghdasi, M. Performance management using a value-based customer-centered model. *Int. J. Prod. Res.* 2015, 53, 5472–5483. [CrossRef]
- 49. Shameem, M.; Kumar, R.R.; Kumar, C.; Chandra, B.; Khan, A.A. Prioritizing challenges of agile process in distributed software development environment using analytic hierarchy process. *J. Softw. Evol. Process* **2018**, *30*, e1979.
- Xu, Y.; Zhou, D.; Wang, W. Being my own gatekeeper, how I tell the fake and the real–Fake news perception between typologies and sources. *Inf. Process. Manag.* 2023, 60, 103228. [CrossRef]
- Pelau, C.; Pop, M.I.; Stanescu, M.; Sanda, G. The Breaking News Effect and Its Impact on the Credibility and Trust in Information Posted on Social Media. *Electronics* 2023, 12, 423. [CrossRef]
- Koch, T.K.; Frischlich, L.; Lermer, E. Effects of fact-checking warning labels and social endorsement cues on climate change fake news credibility and engagement on social media. J. Appl. Soc. Psychol. 2023, 53, 495–507. [CrossRef]
- Masullo, G.M.; Tenenboim, O.; Lu, S. "Toxic atmosphere effect": Uncivil online comments cue negative audience perceptions of news outlet credibility. *Journalism* 2023, 24, 101–119. [CrossRef]
- Aghajari, Z.; Baumer, E.P.; Difranzo, D. Reviewing Interventions to Address Misinformation: The Need to Expand Our Vision Beyond an Individualistic Focus. In *Proceedings of the ACM on Human-Computer Interaction*; Association for Computing Machinery: New York, NY, USA, 2023; Volume 7, pp. 1–34.
- 55. Leeder, C. How college students evaluate and share "fake news" stories. Libr. Inf. Sci. Res. 2019, 41, 100967. [CrossRef]
- 56. Farkas, J.; Neumayer, C. Mimicking news: How the credibility of an established tabloid is used when disseminating racism. *Nord. Rev.* **2020**, *41*, 1–17. [CrossRef]
- Hosni, A.I.E.; Li, K.; Ahmad, S. DARIM: Dynamic approach for rumor influence minimization in online social networks. In Proceedings of the Neural Information Processing: 26th International Conference, ICONIP 2019, Sydney, NSW, Australia, 12–15 December 2019; Proceedings, Part II; Springer: Berlin, Germany, 2019; pp. 619–630.
- Hosni, A.I.E.; Li, K.; Ahmed, S. HISBmodel: A rumor diffusion model based on human individual and social behaviors in online social networks. In Proceedings of the Neural Information Processing: 25th International Conference, ICONIP 2018, Siem Reap, Cambodia, 13–16 December 2018; Proceedings, Part II 25; Springer: Berlin, Germany, 2018; pp. 14–27.
- 59. Kabra, G.; Ramesh, A.; Arshinder, K. Identification and prioritization of coordination barriers in humanitarian supply chain management. *Int. J. Disaster Risk Reduct.* 2015, *13*, 128–138. [CrossRef]
- 60. Albayrak, E.; Erensal, Y.C. Using analytic hierarchy process (AHP) to improve human performance: An application of multiple criteria decision making problem. *J. Intell. Manuf.* **2004**, *15*, 491–503. [CrossRef]
- Chan, F.T.; Kumar, N. Global supplier development considering risk factors using fuzzy extended AHP-based approach. *Omega* 2007, 35, 417–431. [CrossRef]
- 62. Bozbura, F.T.; Beskese, A.; Kahraman, C. Prioritization of human capital measurement indicators using fuzzy AHP. *Expert Syst. Appl.* **2007**, *32*, 1100–1112. [CrossRef]
- 63. Wong, J.K.; Li, H. Application of the analytic hierarchy process (AHP) in multi-criteria analysis of the selection of intelligent building systems. *Build. Environ.* 2008, 43, 108–125. [CrossRef]
- 64. Jan, R.U.; Usman, M.; Abrar, M.F.; Ullah, N.; Asshad, M.; Ali, S. Scaling Agile Adoption Motivators from Management Perspective: An Analytical Hierarchy Process Approach. *Sci. Program.* **2021**, 2021, 4522273. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.